Author's response to reviews

Title: Treatment of Thoracolumbar and Lumbar Curves with P.A.S.B. Brace for Idiopathic Scoliosis: assessment of results following the S.R.S. Commitee on Bracing and Nonoperative Management Standardization Criteria.

Authors:

Angelo G. Aulisa (agaulisa@tiscali.it)
Vincenzo Guzzanti (vguzzanti@yahoo.it)
Marco Galli (dottorgalli@yahoo.it)
Carlo Perisano (carloperisano@hotmail.it)
Francesco Falciglia (falciglia@opbg.net)
Lorenzo Aulisa (aulisa@libero.it)

Version: 4 Date: 15 August 2009

Author's response to reviews:

Dear Prof. TB Grivas

Many thanks for your review,

in the figure 8 and 9 the x-rays at the beginning (t1) and at the end of treatment (t5) are taken without the brace, instead during the treatment the x-rays are with brace because in this way we can evaluate the effectiveness of the brace.

We specified in the text as follows:

“….in t2-t3-t4 the x-rays were carried out with the brace, because in this way we can evaluate the effectiveness of the brace…."

Yours sincerely,
Dr. Angelo Gabriele Aulisa